Author’s response to reviews

Title: An international longitudinal registry of patients with atrial fibrillation at risk of stroke (GARFIELD): the UK protocol

Authors:

Patricia N Apenteng (p.n.k.apenteng@bham.ac.uk)
Ellen T Murray (e.t.murray@bham.ac.uk)
Roger Holder (r.l.holder@bham.ac.uk)
Richard FD Hobbs (richard.hobbs@phc.ox.ac.uk)
David A Fitzmaurice (d.a.fitzmaurice@bham.ac.uk)

Version: 2 Date: 29 November 2012

Author’s response to reviews:

The Competing Interests section has been placed between the conclusion and authors contribution section.

The statement on funding has been moved to the Computing Interests section as the study is supported by a grant from a pharmaceutical company.